%  This template is based on the original example by Athanassios Protopapas, October 2005,
%  mini-example for using apa.cls and modified by William Revelle, August, 2007, available
% here: http://www.personality-project.org/revelle/syllabi/205/apa.style.html

\documentclass[doc]{apa6}%can be jou (for journal), man (manuscript) or doc (document)

%these next packages extend the apa class to allow for including statistical and graphic commands
\usepackage{url}   %this allows us to cite URLs in the text
\usepackage{graphicx}  %allows for graphic to float when doing jou or doc style
\usepackage{amssymb}  %use formatting tools  for math symbols
\usepackage{apacite}
\usepackage{textcomp}
\pagenumbering{gobble}

% type setting of functions, packages, and R follows a particular style
\let\proglang=\textsf
\newcommand{\R}{\proglang{R}}
\newcommand{\pkg}[1]{{\normalfont\fontseries{b}\selectfont #1}}
\newcommand{\Rfunction}[1]{{\texttt{#1}}} 
\newcommand{\fun}[1]{{\texttt{#1}}} 
\newcommand{\Robject}[1]{{\texttt{#1}}} 
\shorttitle{short title}
\bibliographystyle{apacite}

\usepackage[T1]{fontenc}
\usepackage{inconsolata}

\renewcommand{\abstractname}{\vspace{-\baselineskip}}

%Here is where we start the important APA stuff

\title{SYMPOSIUM PROCEEDINGS PAPER TITLE}
\author{Author(s)' Name(s)'}
\affiliation{Author(s)' Affliation(s), City, State (if USA), Country} \\ [1cm]

\abstract{This document provides formatting instructions and serves as a template for ISAP Proceeding papers. The title of the paper should be in all caps and centered at the top of the page. Type the abstract here as a single paragraph in block format (indenting both margins by .5 in or 1.3 cm) and flush left. Do not exceed 150 words. The page limit is 6 pages, including the references, tables, and figures. Use a Times New Roman typeface with a minimum font size 11 for the text. A sans serif typeface may be used in figures. Submit the paper in a pdf file by May 1, 2023 via the ISAP web site, \url{https://www.rit.edu/isap2023/}. Any papers not following these guidelines or received after May 1 will not be included in the Proceedings. If you have any questions, please contact \mailto{esa.rantanen@rit.edu} before the deadline.}
\thispagestyle{empty}
\begin{document}
\maketitle   

\thispagestyle{empty}

Neither the Abstract nor the Introduction should have a heading. The paper should fit on a 8 1/2 $\times$ 11 in page. Leave uniform margins of 1 in (2.54 cm) at the top, bottom, left, and right of every page. Use single space except between paragraphs and headings throughout the paper. Use the flush-left style (with indented first lines of paragraphs) and do not right-justify lines. 
 
Use the format of this document as a template for your paper. Your following the format diligently and consistently will result in a much more professional-looking archive of our work. Students should have their adviser or a senior colleague review their papers before submitting. Take care to proofread your paper before submitting because all papers will be published as is. 

\section{Detailed Instructions (Level 1 Heading)}

To provide a consistent format for the proceedings, please type your paper using the format as shown in these instructions. It should be carefully proofed for spelling and format errors. Note that authors are responsible for securing permission to reproduce figures or text from authors and publishers concerned. 

\subsection{Levels of Heading (Level 2 Heading)}

The five levels of headings are: Level 1 -- \textbf{Centered, Boldface, Uppercase and Lowercase Heading}; Level 2 -- \textbf{Flush Left, Boldface, Uppercase and Lowercase Heading}; Level 3 -- \textbf{Indented boldface, lowercase paragraph heading ending with a period}; Level 4 -- \textbf{\textit{Indented, boldface, italicized, lowercase paragraph heading ending with a period}}; Level 5 -- \textit{Indented, italicized, lowercase paragraph ending with a period}. For a paper that only has one level of heading, use Level 1; for a paper with two levels of heading, use Levels 1 and 2; if three levels are needed, use Levels 1, 2, and 3; and so forth. 

\subsection{Tables and Figures (Level 2 Heading)}

\subsubsection{Table and table title (Level 3 heading)}
 Number all tables with Arabic numerals in the order in which they are first mentioned in text. In the text, refer to tables by their number (e.g., ``as shown in Table~\ref{tab:pastISAP}'').     

\begin{table}[h!]
\caption{\\Previous Locations of the International Symposium of Aviation Psychology.} 
\begin{center}
\begin{tabular}{l l}
\hline
Year 		&	Location	\\
\hline
2001		&	Columbus, Ohio	\\
2003		&	Dayton, Ohio		\\
2005	 	&	Oklahoma City, Oklahoma		\\
2007		&	Dayton, Ohio	\\
2009		&	Dayton, Ohio	\\
2011		&	Dayton, Ohio	\\
2013		&	Dayton, Ohio	\\
\hline	\\
\end{tabular}
\end{center}
\textit{Note}. Table note goes to the bottom of the table. Only the word ``Note'' is italicized.
\label{tab:pastISAP}
\end{table}
          
\subsubsection{Figure and figure caption (Level 3 heading)}
 Number all figures with Arabic numerals in the order in which they are first mentioned in the text. Place the figure caption below the figure. Italicize the word ``Figure'' and the appropriate number, followed by a period, flush left. Type the caption in a sentence form without italicizing. The caption serves both as an explanation of the figure and as a figure title. It should summarize the main points portrayed in the figure. Insert tables and figures where they are first mentioned in the text and not at the end of the paper.

\subsection{References (Level 2 Heading)}

Follow the APA (7th edition) format. References are cited in the text with an author-date citation, for example, \citeA{adams1979evaluation} or \cite{adams1979evaluation}, and are listed alphabetically in the reference list at the end of the paper. 

For the reference list, center the word ``References'' typed in uppercase and lowercase letters. List references alphabetically by first author. Use a hanging indent format with the first line of each reference set flush left and subsequent lines indented. Cite journals in full and include the volume and page numbers. Include the digital object identifier (DOI) in the reference list if one is assigned. For articles retrieved online, include the homepage URL. Use this format: ``Retrieved from http://www.xxxxxxxxxx.''  Please take the time to furnish complete references to help the readers to locate the original work.

Examples of different sources of references and their full citation formats are given in the reference list below, including journals \cite{adams1979evaluation}, technical reports \cite{willems1999air}, books \cite{buck1994pilot}, chapters in edited books \cite{durso201820}, and conference proceedings \cite{fothergill2008effect}. 

\section{Acknowledgements (Level 1 Heading)}

Identify grants or other financial support for the study. Acknowledge colleagues who assisted in conducting the study or critiquing the paper. Disclose any special circumstances such as the paper being based on an earlier study, a doctoral dissertation, or variation of related research. If any relationships may be perceived as a conflict of interest, explain them here. Also include any required disclaimer such as stating that the views of the research reported does not reflect the views of the employers or granting organization. 

\bibliography{Refs.bib}

\end{document}